Author's response to reviews

Title: An Intervention Program to Reduce the Number of Hospitalizations of Elderly Patients in a Primary Care Clinic

Authors:

Roni Peleg (pelegr@bgumail.bgu.ac.il)
Yan Press (yanp@zahav.net.il)
Maya Asher (mayas@clalit.org.il)
Tatiana Pugachev (pelegr@bgu.ac.il)
Hadas Glicensztain (pelegr@clalit.org.il)
Mila Lederman (ronipel@gmail.com)
Aya Biderman (sbider@netvision.net.il)

Version: 2 Date: 18 August 2007

Author's response to reviews: see over
August 17, 2007

Editorial Office, BMCHSR

Enclosed please find a paper entitled “An intervention program to reduce hospitalizations of elderly patients in a primary care clinic” that we are submitting for consideration for publication in your journal. The results of this study show that implementation of the intervention program in a single primary care clinic led to a reduction in hospitalizations for the elderly patient population and to a more optimal allocation of healthcare resources.

Thank you for your consideration.

Respectfully,

Roni Peleg, MD